Massachusetts Institute of Technology Handout 5 6.915: Network and Computer Security September 21, 1995 Professor Ronald L. Rivest

Readings for Lecture 5

Attached is the following list of papers

ffl How to Share a Secret by Adi Shamir [3] ffl Toward a National Public Key Infrastructure by Santosh Chokhani [1] ffl Key Escrowing Today by Dorothy Denning and Miles Smid [2]

References

[1] Santosh Chokhani. Toward a national public key infrastructure. IEEE Communications Magazine, 32(9):70{74, September 1994.

[2] Dorothy E. Denning. Key escrowing today. IEEE Communications Magazine, 32(9):58{ 68, September 1994.

[3] A. Shamir. How to share a secret. Communications of the ACM, 22:612{613, November 1979.